Author's response to reviews

Title: A case of non-small cell carcinoma lung with choroidal metastasis as presenting sign, showing good response to chemotherapy alone.

Authors:

Preety Shukla (drpreetyshukla_08@rediffmail.com)  
Abhishek Singh (hivneg@gmail.com)  
Parul Singh (parulophtha@gmail.com)  
Vikas shukla (vikasshukla_08@rediffmail.com)

Version: 2 Date: 26 March 2008

Author's response to reviews:

Respected Sir,

As per your kind instructions, following changes have been done in the manuscript-

1- Title restructured so it ends in ‘A presenting with B in C: a case report’
2- Conclusion section already present in the manuscript.
3- Keywords and key message sections removed from the abstract.
4- Title removed from page 2.

Thanking you!!

Yours Sincerely,

Abhishek Singh